Author’s response to reviews

Title: Ileal intubation is not associated with higher detection rate of right-sided conventional adenomas and serrated polyps compared to cecal intubation after adjustment for overall adenoma detection rate

Authors:

Martin Buerger (martin.buerger@uk-koeln.de)

Philipp Kasper (philipp.kasper@uk-koeln.de)

Gabriel Allo (gabriel.allo@uk-koeln.de)

Johannes Gillessen (johannes.gillessen@uk-koeln.de)

Christoph Schramm (christoph.schramm@uk-koeln.de)

Version: 3 Date: 03 Nov 2019

Author’s response to reviews:

Reply to Editor comment No. 1:

Because all identifying Information has been appropriately anonymised and there are no identifying Images used within the manuscript, consent to publish is not applicable to our manuscript. "Not applicable" was added to this point.

Reply to Editor comment No. 2:

Declarations section at the beginning of the manuscript was removed.

Reply to Editor comment No. 3:

The manuscript was cleaned from any tracked changes, comments, Highlights, strikethroughs and text in different Colours.